RESEARCH Open Access



Relationship between eating disorders perception and psychosocial profile in school-dropout adolescents

Francesca Mastorci¹, Maria Francesca Lodovica Lazzeri¹, Paolo Piaggi², Cristina Doveri¹, Anselmo Casu¹, Gabriele Trivellini¹, Irene Marinaro¹, Cristina Vassalle³ and Alessandro Pingitore^{1*}

Abstract

Background In body-mind relationship field, eating disorders (ED) are considered disabling disorders that can alter physical health status, inducing profound alterations in psychosocial, cognitive and emotional dimensions. These disorders, characterized by a strong comorbidity with other diseases, usually begin during childhood or adolescence, and include anorexia nervosa, bulimia nervosa and binge eating. Aim of this study was to investigate the associations between eating disorders perception and dimensions of health-related quality of life (HRQoL) and well-being perception (WBP) in school-dropout adolescents.

Methods Data were collected in 450 adolescents (19 \pm 2 years, male 308), and HRQoL, WBP, and ED were assessed by means a battery of standardized questionnaire.

Results EDs are more pronounced in females than in males (p < 0.05) and are associated with lower HRQoL (p < 0.001) and lower well-being perception (p < 0.001). EDs are associated with an impairment of physical (p < 0.05) and psychological well-being perception (p < 0.001), emotional responses (p < 0.001), self-perception (p < 0.001), and a reduction of general well-being (p < 0.05).

Conclusions Although it is complicated to distinguish between causes and consequences, these findings suggest a complex and multifaceted, association between ED and HRQoL domains. Thus, multiple factors need taking into account in the policy of EDs prevention, identifying all the components of well-being to focus and personalize healthy programs in adolescences.

Keywords Health-related quality of life, Adolescence, Eating disorders, Body perception, Well-being

Plain English summary

Adolescence is a transitional period from childhood to adulthood during which important biological, cognitive and social changes are experienced. In this phase, characterized by increased psychological vulnerability, classic risk behavior can develop, including altered eating behavior leading to eating disorders (EDs). EDs represent a varied group of pathological conditions influenced by various individual factors, including emotional state, cognitive functioning, and social context, probably associated to physical, emotional, and social changes that in young people can lead to great instability. Our results shown that is complicated to distinguish between causes and consequences

*Correspondence: Alessandro Pingitore pingi@ifc.cnr.it Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons locence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

that cause EDs, suggesting a complex and multifaceted association between ED and psychological domains. This is important in order to develop preventive and early intervention programs in the schools and in other educational institutions, focusing on what are considered potential predictor of EDs.

Introduction

Adolescence is a period of transition from childhood to adulthood in which people undergo to many changes in almost areas of their life, including biological dimensions, cognitive functions, and social relationships [1]. Although most adolescents overcome these challenges successfully, adolescence is a time of increased risk behaviour, so much so that more than a third of adult psychopathologies find their onset during adolescence [2]. The possibility that particular risk factors for some developing young people may represent normal experiences, while for others they are indicators of a fragility that becomes a pathology over time, makes understanding psychological development in this time window particularly important [2]. In adolescence, within the body-mind relationship and especially in body-focused disorders, eating disorders (EDs) represent a varied group of pathological conditions influenced by various individual factors, including emotional state, cognitive functioning, and social context, probably associated to physical, emotional, and social changes that in young people can lead to great instability [3–7]. Despite the heterogeneity of the clinical and psychological condition, according the American Psychiatric Association, the three most common EDs are anorexia nervosa, bulimia nervosa, and binge eating disorder [8]. Currently epidemiological data indicate an increase in EDs, with a higher prevalence in female population as compared to males and frequent psychiatric comorbidities [9-12]. One of the most significant indicators is self-esteem that usually correlates with health and well-being [13-15]. In particular, positive self-esteem promotes better social adaptation and reduces the occurrence of risk behaviour [16]. Especially in adolescence, self-esteem is related to body image, more properly to body appreciation, considered a mediating factor for EDs. Body appreciation is a psychophysical construct that, when positive, is relates to healthy eating behaviour and good social relations, while is negatively associated with dysfunctional behaviors such as alcohol and drug use [17]. This gives insight into the complexity of EDs, but at the same time opens up the need to assess how EDs are associated with different health-related variables, not only the more purely physical ones, but above all those of an emotional and social nature. Among these variables impacting on EDs is certainly the phenomenon of dropping out of school, which to our knowledge has been little studied in acute cases, but is correlated in adulthood with a series of dysfunctional behaviours. Considering the impact of dropout on long-term healthrelated variables, the aim of this study was to assess the relationship between several dimensions of self-reported health-quality of life (HRQoL) and well-being perception with EDs in a sample of Italian school-dropout adolescents, with focus on possible sex differences.

Materials and methods

Participants

Data from school dropout were collected as part of AVA-TAR project, acronym for "A new purpose for promotion and eVAluation of healTh and well-being Among healthy teenageRs". AVATAR project is aimed to develop a new tool to assess lifestyle habits, social context, and emotional status, in adolescents, and to define an integrated index of the best indicators of well-being [18]. In total, 680 boys and girls, aged between 18 and 21 years, were included. Adolescents who dropped out of school were enrolled in collaboration with the regional agency FOR-MATICA to which they were enrolled after dropping out for a professional training. School dropout adolescents were referred to the Tuscany Region's FORMATICA Agency for a 2-year professional training course. This agency deals precisely with recovering those adolescents who have voluntarily interrupted their studies, offering them a training opportunity for a better integration into the professional world. The inclusion criteria were the following: absence of neuropsychiatric or other pathologies, signed informed consent and full completion of the proposed questionnaires.

Of 680 adolescents initially considered, 230 were excluded for the following reasons: diagnosed neuropsychiatric or other diseases (n=20), absence of sign informed consent (n=75), questionnaires not filled completely (n=135). Therefore, the final population consisted of 450 adolescents (19 ± 2 years, male 308). Participants were previously instructed on how to fill out the questionnaires and to conduct the tests. One or two project members visited professional training course to provide the adolescents with verbal and written information about the data collection. Written information was also distributed to the parents. Active informed consent was obtained from adolescents. All tests were performed during participants' computer lesson in school time. No incentive was provided to adolescents or parents. A research assistant was available to provide information and technical support to complete questionnaires.

Ethical statement

The research protocol was approved by the local ethics committee review board. All subjects or legal guardians gave informed consent, and authorized physicians to use their clinical data in accordance with the Italian law. All procedures performed in the study were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki declaration and its later amendments or comparable ethical standards (Regional Ethics Committee for Clinical Trials of the Tuscany Region 36/2021).

Data collection

Data were collected with AVATAR Web-tool [18]. A socio-demographic data record was used to collect information on gender, age, and schooling. The Italian version of KIDSCREEN-52 was used to assess health-related quality of life [19, 20]. The KIDSCREEN is a self-report questionnaire designed to address health-related quality of life (HRQoL), aimed to monitor and measure the personal experiences in children and adolescent about their perception of health status and well-being. The questionnaire, that describes physical, psychological, mental, social, and functional aspects of well-being, consists of 52 items grouped in 10 dimensions (Physical well-being, Psychological wellbeing, Moods and emotions, Self-perception, Autonomy, Parent relations and home life, Social support and peers, School environment, Social acceptance (bullying), Financial resources) [19, 20]. Some sample items "In general, how would you say your health is?" for Physical well-being dimension; "Have you felt satisfied with your life?" for Moods and emotions; "Have you been happy with the way you are?" for Self-perception. Cronbach's alphas are ranging from 0.77 to 0.89 for the dimensions of the 52-item version. KIDSCREEN questionnaires is psychometrically tested using data obtained in a multicentre European study which included a sample of 22,827 children recruited in 13 countries [21]. Except for mood and bullying, higher values of the variables express a better health-related quality of life. KIDSCREEN questionnaires is psychometrically tested using data obtained in a multicentre European study which included a sample of 22,827 children recruited in 13 countries. Eating attitudes and behaviours were assessed using EAT-26 consisting of three subscales: Dieting, eating preoccupation and oral control [22]. The EAT-26 consists of three sections: (a) self-reported height and weight to create a body mass index (BMI), (b) 26 items rated on a six-point Likert scale related to how often an individual engages in certain behaviors ("Always," "Usually," "Often," "Sometimes," "Rarely," and "Never"), and (c) five behavioural items on a six-point Likert scale examining how often a person has engaged in disordered eating behaviors over the past 6 months ("Never," "Once a month or less," "2–3 times a month," "Once a week," "2–6 times a week," and "Once a day or more").

Personalized well-being index

The AVATAR approach consists in focusing on the integration of three components of health-related well-being [lifestyle habits (LH); emotional status (ES); and social context (SC)], as perceived by adolescent [23]. The three components were obtained from the different variables analyzed by the questionnaires according to a structural model previously described in Mastorci et al. [24].

In detail, the path analysis technique used measures the extent to which the model fits a data set and allows testing of interrelationships between several variables simultaneously. The confirmatory factor analysis was used to test an overall measurement model that included five correlated latent variables. Overall model fit was assessed using different statistics. First, a chi-square analysis was used. The other indices were the root mean square error of approximation (RMSEA) (values between 0.05 and 0.08 indicate acceptable fit, and values < 0.05 a good fit), comparative fit index (CFI) (values > 0.90 indicate reasonable fit, > 0.95 good fit), and standardized root mean square residual (SRMR) (values < 0.10 indicate good fit). The measurement model was first tested to ensure that each of the observed variables was a sufficient indicator of the hypothesized latent variables.

From the sum of the three components, we obtained a personalized well-being index (PWBI), ranging from 0 to 100, according to the AVATAR model as reported before [24].

Statistical analysis

Statistical data analyses were performed using SPSS software. Data are presented as mean \pm SD, while categorical variables were presented as counts and percentages. The Saphiro–Wilk test was used to confirm the normality of data distribution for continuous variables before parametric analyses. A p value \leq 0.05 was considered statistically significant.

One-way between-groups multivariate analyses of variance were performed to identify if EDs are associated with the dimensions of health-related quality of life and PWBI. These multivariate analyses were based on Wilks' Lambda statistic after Bonferroni adjustment of significance for multiple tests based on 14 risk behaviors tested and on gender. In case of significance by the multivariate analysis, post-hoc analyses were conducted to explore inter-group differences for each dimension of health-related quality of life and PWBI.

Results

Association between eating disorders, health-related quality of life and personalized well-being index in study population and by gender

In total, 450 participants (32% girls, mean age 19.38 ± 1.89) were included in the analyses. Age was similar between male and female (male 19.35 ± 1.69 vs female 19.46 ± 2.28 , p = ns).

Scores higher for risk of EDs were observed in the female population as compare to counterparts (14% vs 5.8%, p<0.01). In total population, after adjustment for multiple tests, eating disorders was significantly associated with HRQoL as total score (Wilks' Lambda=0.87, adj. p<0.001). In particular, considering the single components of HRQoL, EDs were related to an impairment in physical (F=9.289; p=0.025) and psychological well-being perception (F=26.964; p<0.001), mood (F=23.079; p<0.001), and self-perception (F=18.927; p<0.001). After adjustment of HRQoL by gender, association between EDs and quality of life is maintained indicating that it is not due to gender (F=14.220; p<0.001).

With respect to well-being perception, assessed through PWBI, after adjustment for multiple tests, eating disorders in total population were significantly associated with PWBI (Wilks' Lambda = 0.89, adj. p < 0.001), in terms of reduction of score (F=22.85; p < 0.001). Taking into account the three components of the index, EDs correlated with greater criticality of emotional status (F=39.64; p < 0.001). After adjustment of PWBI by gender, association between EDs and well-being perception is preserved showing that it is not due to gender (F=15.08; p < 0.001).

Discussion

The first objective of this study was to identify the relationship between health-quality of life in its dimensions and well-being perception in its components with eating disorders perception in a sample of Italian school-dropout adolescents. Findings indicate an overall association between eating disorders and health-related quality of life and well-being perception in terms of total score. More specifically the presence of EDs is related to an impairment of physical and psychological well-being, mood and self-perception. The second objective was to analyse whether these relationships were sex-dependent, indicating that although there was a higher prevalence of EDs in the female population, this difference in the relationship with the dimensions related to well-being was not maintained. In other words, both in male and female adolescents, EDs negatively impacts on psychosocial dimensions.

Eating disorders are a category of disease often underdiagnosed and many adolescents go untreated, do not recover or recover only partially. High rates occur in younger children, adolescents and minority groups, presenting themselves differently in these populations, which is why recognizing them early is not always easy, although to our knowledge there is no data on EDs among our study sample, school-dropout adolescents [7]. In line with previous data, a relationship between self-perception and EDs was shown [25]. Our results are consistent with that reported by prior research, revealing the presence of EDs when self-perception is low [26]. Of all health-related dimensions, in facts, self-perception in adolescence appears to be the most important, even playing the role of protective or preventive factor against EDs [27]. Indeed, it is common that low self-perception co-occurs with other psychological traits of EDs creating a vicious cycle. On the other hand, our results demonstrated also a relation with an impairment in physical and psychological well-being perception and reduced mood level, suggesting a compromised emotional profile. Generally, during adolescence, these constructs may be associated with the definition of one's body image. Traditionally, body image and health are considered associated only in pathological conditions, mainly because a negative body perception is thought as a risk factor for eating disorders and other psychological diseases.

In this frame, as suggested by our results, there is another important aspect concerning the association between the perception of body size and mental health, and subsequent impact on EDs [28]. In a previous study, our group demonstrated that, in adolescent, exists a closed relationship between weight status categories and HRQoL, especially in girls and particularly in underweight girls, who presented a greater perception of well-being, pointing out how this in the long term can contribute to psychological distress that can lead to an eating disorder [29]. This is in line with our notion that young female usually reports aim to weight loss, whereas a higher percentage of boys report the goal to maintain their weight. With regard to sex differences and the possible different association between EDs, HRQoL, and PWBI, our results have shown that these relationships are present equally both in male and in female students, although the prevalence of EDs was higher in girls as expression of a biological predisposition, although female population is composed only by 32% [30]. These findings are probably due to the fact that EDs are affected by cultural standards related to the body as a social fact rather than a biological one [31]. Usually, the predominance of research on eating disorders, also in terms of prevention, focuses mainly on females, although in recent years the percentage of males with these disorders has increased

significantly, with onset beginning at around age 11 [32]. Although also the data on the co-occurring psychological conditions are more described in females, some studies reported male with EDs more likely to have depression, even if they are still inconsistent data [33]. In general, the inconsistency of data on possible sex differences and related co-morbidities that may somehow exacerbate EDs, may be due on the different weight control behaviors; boys are potentially focusing on developing muscularity, while females on leanness [34]. In fact, in the male gender, factors such as body image related to muscularity and sexual orientation may increase the risk of developing eating disorders [35]. In young women, on the other hand, the beauty ideals proposed by social media seem to have a great influence. Indeed, there is a relationship between body image, body concerns, body dissatisfaction and disordered eating attitudes in the female gender [36]. In particular, in girls has been described an association between time spent on social networks and more significant eating disorders symptoms [37].

Notable possible strengths of this study include the possibility to develop preventive strategies on psychosocial dimensions in order to reduce the vulnerability and the incidence of the ever-increasing EDs in adolescent. In addition, it is important to emphasize that this is a first pilot study conducted on school-dropout adolescents, who are therefore potentially more exposed to risk factors, as the little literature aimed at this adolescent population group indicates. A recent work of our group revealed that school-dropout girls are characterized by a compromised health indicator profile in terms of both risk behaviors, including EDs [38]. The study of the dropout phenomenon is new, as most of the evidence focuses on the long-term effects of dropping out of school, but there are no studies to our knowledge on the short- and medium-term effects. School dropout opens up the possibility of developing risky behavior and vulnerability to disease in adulthood. So, it is like having an increased susceptibility, in a biological period already of increased fragility as adolescence.

Although the current study provides the groundwork for future research, certain limitations warrant discussion. First, the sample consisted of 450 dropout adolescents between the ages of 18 and 21 and identified as predominately of North Italy, are not representative of overall Italian Country neither of school-dropout students. Also, the relatively small sample size may have limited power to detect smaller effect sizes, especially as it relates to EDs, HRQoL categories, and PWBI. Future studies on a larger and more representative population will be necessary to confirm our preliminary results. Furthermore, in light of the results of our previous research on the link between body weight and quality of life, it

would also have been important to include body weight. However, this was not done as it was self-assessed and may not be a real result. Considering the role of social networks in the onset of EDs, in future studies on the relationship between mental functioning and development of body-focused disorders, it will also be important to consider this aspect. Furthermore, a more accurate association between HRQoL and EDs could be studied using the recently validated Italian version of the Eating Disorder Quality of Life [39]. Another important aspect to assess is sexual orientations, because, especially in the analysed age, has a serious role in the EDs onset and psychopathology.

Lastly, more important information about EDs and psychosocial profile would be obtained if they compared those who dropped out to those who did not drop out.

Conclusion

The current study demonstrates that there is an association between EDs and health-related quality of life and well-being perception; in other terms when the disorder is present are impaired physical and psychological wellbeing, mood and self-perception. However, males and females reported the same level of these associations, in alignment with some previous reports. Long-term monitoring is needed to evaluate whether these impairments persist over time and how they affect the psychosocial sphere in terms of co-morbidity. Our results are important because they refer to an at-risk cohort although obtained on a small pilot sample, they shed light on a complex and multifaceted phenomenon such as schooldropout. In this sense, it is as if in our sample coexisted psychological, physical, and social alterations typical of EDs and the same changes characteristic of school-dropout, without however knowing who causes what. Future research elucidating the nature of EDs in school-dropout adolescents, and how the alterations here presented may differ from regular students, is warranted. Additional research establishing the psychosocial features of EDs symptom in school-dropout adolescents is also required, in order to develop preventive and or early intervention programs in eating disorders. Promoting health has long been a pivotal role of schools and of other educational institutions, but traditionally these activities have focused mainly on general health education. In this context, according our results, it would be useful to focus on self-esteem, which not surprisingly is impaired in schooldropout and is reputed to be a predictor of EDs.

Abbreviations

ED Eating disorders
HRQoL Health-related quality of life
WBP Well-being perception

AVATAR A new purpose for promotion and eVAluation of healTh and well-

being Among healthy teenageRs

LH Lifestyle habits
ES Emotional status
SC Social context

Acknowledgements

Not applicable.

Author contributions

FM conceptualization, writing-original draft, methodology, review and editing; MFLL writing-original draft, methodology, review and editing; PP software, formal analysis, review and editing; CD methodology, software, data acquisition; AC methodology, software, data acquisition; IM data acquisition; CV conceptualization, review and editing; AP conceptualization, writing-original draft, methodology, review and editing, supervision. All authors read and approved the final manuscript.

Funding

No funding to declare.

Availability of data and materials

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

Declarations

Ethics approval and consent to participate

Regional Ethics Committee for Clinical Trials of the Tuscany Region 36/2021.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Clinical Physiology Institute, CNR, Via Moruzzi, 1, 56124 Pisa, Italy. ²Department of Information Engineering, University of Pisa, Pisa, Italy. ³Fondazione Toscana Gabriele Monasterio, Pisa, Italy.

Received: 3 November 2022 Accepted: 29 March 2023 Published online: 11 April 2023

References

- Patton GC, Coffey C, Cappa C, Currie D, Riley L, Gore F, Degenhardt L, Richardson D, Astone N, Sangowawa AO, et al. Health of the world's adolescents: a synthesis of internationally comparable data. Lancet. 2012;379:1665–75.
- Casey BJ, Jones RM, Hare TA. The adolescent brain. Ann N Y Acad Sci. 2008;1124:111–26. https://doi.org/10.1196/annals.1440.010. (PMID: 18400927; PMCID: PMC2475802).
- Marcus MD, Wildes JE. Obesity: is it a mental disorder? Int J Eat Disord. 2009;42:739–53.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 5th ed. Washington: American Psychiatric Association; 2013.
- Zipfel S, Giel KE, Bulik CM, Hay P, Schmidt U. Anorexia nervosa: aetiology, assessment, and treatment. Lancet Psychiatry. 2015;2:1099–111.
- Treasure J, Zipfel S, Micali N, et al. Anorexia nervosa. Nat Rev Dis Primers. 2015;1:15074.
- Campbell K, Peebles R. Eating disorders in children and adolescents: state of the art review. Pediatrics. 2014;134(3):582–92. https://doi.org/ 10.1542/peds.2014-0194. (PMID: 25157017).
- 8. American Psychiatric Association; 2021.

- Galmiche M, Déchelotte P, Lambert G, Tavolacci MP. Prevalence of eating disorders over the 2000–2018 period: a systematic literature review. Am J Clin Nutr. 2019;109(5):1402–13. https://doi.org/10.1093/ajcn/nqy342.
- Treasure J, Duarte TA, Schmidt U. Eating disorders. Lancet. 2020;395(10227):899–911. https://doi.org/10.1016/S0140-6736(20)30059-3. (PMID: 32171414).
- Keski-Rahkonen A, Mustelin L. Epidemiology of eating disorders in Europe: prevalence, incidence, comorbidity, course, consequences, and risk factors. Curr Opin Psychiatry. 2016;29:340–5.
- 12. Udo T, Grilo CM. Psychiatric and medical correlates of DSM-5 eating disorders in a nationally representative sample of adults in the United States. Int J Eat Disord. 2019;52:42–50.
- Bąk-Sosnowska M, Gruszczyńska M, Skrypnik D, Grzegorczyn S, Karolkiewicz J, Ratajczak M, Mądry E, Walkowiak J, Bogdański P. Type of physical training and selected aspects of psychological functioning of women with obesity: a randomised trial. Nutrients. 2021;13:2555. https://doi.org/ 10.3390/nu13082555.
- Brechan I, Kvalem IL. Relationship between body dissatisfaction and disordered eating: mediating role of self-esteem and depression. Eat Behav. 2015;17:49–58. https://doi.org/10.1016/j.eatbeh.2014.12.008.
- Unlu S, Aykut M, Borlu A, Kaner G. Are low self-esteem and body image dissatisfaction related with body mass index? Prog Nutr. 2019;21(1-S):94–103.
- Reina MC. Percepciones de autoevaluación: Autoestima, autoeficacia y satisfacción vital en la adolescencia. Psychol Soc Educ. 2010;2(1):55–69.
- Nolen E, Panisch LS. The relationship between body appreciation and health behaviors among women and adolescent girls: a scoping review. Health Soc Work. 2022;47(2):113–22.
- Trivellini G, Doveri C, Mastorci F, Bastiani L, Cappa C, Vassalle C, et al. Innovative web-based tool for promoting well-being among healthy adolescents: an implementation protocol. J Transl Sci. 2018;5:1–5.
- Ravens-Sieberer U, Gosch A, Rajmil L, Erhart M, Bruil J, Duer W, Auquier P, Power M, Abel T, Czemy L, Mazur J, Czimbalmos A, Tountas Y, Hagquist C, Kilroe J, Kidscreen Group E. KIDSCREEN-52 quality-of-life measure for children and adolescents. Expert Rev Pharmacoecon Outcomes Res. 2005;5(3):353–64. https://doi.org/10.1586/14737167.5.3.353. (PMID: 19807604).
- The KIDSCREEN Group Europe. The KIDSCREEN questionnaires—quality
 of life questionnaires for children and adolescents. Handbook; Pabst Science Publishers: Lengerich, Germany. 2006.
- Berra S, Ravens-Sieberer U, Erhart M, Tebé C, Bisegger C, Duer W, von Rueden U, Herdman M, Alonso J, Rajmil L, European KIDSCREEN group [kidscreen]. Methods and representativeness of a European survey in children and adolescents: the KIDSCREEN study. BMC Public Health. 2007;7:182. https://doi.org/10.1186/1471-2458-7-182. PMID: 17655756; PMCID: PMC1976616.
- 22. Garner D, Olmsted M, Bohr Y, Garfinkel P. The eating attitudes test: psychometric features and clinical correlates. Psychol Med. 1982;12:871–8.
- Mastorci F, Bastiani L, Trivellini G, Doveri C, Vassalle C, Pingitore A. A new integrated approach for adolescent health and well-being: the AVATAR project. Health Qual Life Outcomes. 2020;18:1–9. https://doi.org/10.1186/ s12955-020-01291-6.
- Mastorci F, Bastiani L, Doveri C, Trivellini G, Casu A, Vassalle C, Pingitore A. Adolescent health: a framework for developing an innovative personalized well-being index. Front Pediatr. 2020. https://doi.org/10.3389/fped. 2020.00181.
- Jordana-Ovejero O, Espinoza Guzmán P, González M, Subiza Pérez I, Becerra Castro A, Raich Escursell RM, Mora Giral MS. Universal prevention program of eating, weight and body image problems in adolescents: a 12-month follow-up. Psicothema. 2020;32(2):204–13.
- van Noort BM, Lohmar SK, Pfeiffer E, Lehmkuhl U, Winter SM, Kappel V. Clinical characteristics of early onset anorexia nervosa. Eur Eat Disord Rev. 2018;26(5):519–25.
- Argyrides M, Anastasiades E, Alexiou E. Risk and protective factors of disordered eating in adolescents based on gender and body mass index. Int J Environ Res Public Health. 2020;17(24):9238.
- Whitehead RD, Cosma A, Cecil J, Currie C, Currie D, Neville F, Inchley J. Trends in the perceived body size of adolescent males and females in Scotland, 1990–2014: changing associations with mental well-being. Int J Public Health. 2018;63(1):69–80. https://doi.org/10.1007/s00038-017-0997-y. Epub 2017 Jul 1. PMID: 28668973; PMCID: PMC5766710.

- Mastorci F, Piaggi P, Doveri C, Trivellini G, Marinaro I, Casu A, Pingitore A, Vassalle C. Relationship between weight status and health-related quality of life in a sample of early adolescents from central and northern Italy: a cross-sectional study of the AVATAR project participants. Int J Environ Res Public Health. 2021;18(16):8782. https://doi.org/10.3390/ijerph18168782. (PMID: 34444531; PMCID: PMC8392570).
- Culbert KM, Sisk CL, Klump KL. A narrative review of sex differences in eating disorders: is there a biological basis? Clin Ther. 2021;43(1):95–111. https://doi.org/10.1016/j.clinthera.2020.12.003. Epub 2020 Dec 26. PMID: 33375999: PMCID: PMC7902379.
- Viner RM, Haines MM, Taylor SJ, Head J, Booy R, Stansfeld S. Body mass, weight control behaviours, weight perception and emotional wellbeing in a multi-ethnic sample of early adolescents. Int J Obes. 2006;30:1514.
- Jaworski M, Panczyk M, Śliwczyński A, Brzozowska M, Janaszek K, Małkowski P, Gotlib J. Eating disorders in males: an 8-year population-based observational study. Am J Men's Health. 2019;13(4):1557988319860970.
- Ridout SJ, Ridout KK, Kole J, Fitzgerald KL, Donaldson AA, Alverson B. Comparison of eating disorder characteristics and depression comorbidity in adolescent males and females: an observational study. Psychiatry Res. 2021;296:113–6.
- Chu J, Ganson KT, Vittinghoff E, Mitchison D, Hay P, Tabler J, Murray SB, Nagata JM. Weight goals, disordered eating behaviors, and BMI trajectories in US young adults. J Gen Intern Med. 2021;36:1–9.
- Gorrell S, Murray SB. Eating disorders in males. Child Adolesc Psychiatr Clin N Am. 2019;28(4):641–651. https://doi.org/10.1016/j.chc.2019.05.012. Epub 2019 Jul 11. PMID: 31443881; PMCID: PMC6785984.
- Aparicio-Martinez P, Perea-Moreno AJ, Martinez-Jimenez MP, Redel-Macías MD, Pagliari C, Vaquero-Abellan M. Social media, thin-ideal, body dissatisfaction and disordered eating attitudes: an exploratory analysis.
 Int J Environ Res Public Health. 2019;16(21):4177. https://doi.org/10.3390/ijerph16214177. (PMID: 31671857; PMCID: PMC6861923).
- Wilksch SM, O'Shea A, Ho P, Byrne S, Wade TD. The relationship between social media use and disordered eating in young adolescents. Int J Eat Disord. 2020;53(1):96–106. https://doi.org/10.1002/eat.23198. (Epub 2019 Dec 3 PMID: 31797420).
- Mastorci F, Lazzeri MFL, Piaggi P, Doveri C, Casu A, Trivellini G, Marinaro I, Bardelli A, Pingitore A. Gender differences for health indicators in a sample of school dropout adolescents: a pilot study. Int J Environ Res Public Health. 2022;19(13):7852.
- Meneguzzo P, Todisco P, Calonaci S, Mancini C, Dal Brun D, Collantoni E, Donini LM, Tenconi E, Favaro A. Health-related quality of life assessment in eating disorders: adjustment and validation of a specific scale with the inclusion of an interpersonal domain. Eat Weight Disord. 2021;26(7):2251–62. https://doi.org/10.1007/s40519-020-01081-5.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

